Author's response to reviews

Title: Use of cultivation-dependent and -independent techniques to assess contamination of central venous catheters: a pilot study

Authors:

Mette KS Larsen (mkslarsen@gmail.com)
Trine R Thomsen (trt@bio.aau.dk)
Claus Moser (moser@dadlnet.dk)
Niels Høiby (hoiby@inet.uni2.dk)
Per H Nielsen (phn@bio.aau.dk)

Version: 7 Date: 17 October 2008

Author's response to reviews:

Dear Editor

I have now made the requested changes to the manuscript, and I hope this is OK. The following has been corrected:

The list of authors in the manuscript are now written exactly as they are in the submission system.

The abstract has been structured into Background, Methods, Results, Conclusions.

A conclusions section has been included after the discussion.

Line numbers are removed.

Country is now included in all affiliation details.

Running title has been removed.

Email addresses are included for all the authors on the title page.

Materials & Methods - has been renamed to 'Methods'.

The table legends and the tables themselves are now on separate pages.

Figure 2 has been cropped.

Please let me know if this is not sufficient.

Thanks and all the best from Trine